Author’s response to reviews

Title: Vedolizumab Trough Level Monitoring In Inflammatory Bowel Disease: A State Of The Art Overview

Authors:
Lieven Pouillon (lievenpouillon@icloud.com)
Séverine Vermeire (severine.vermeire@uzleuven.be)
Peter Bossuyt (peter.bossuyt@imelda.be)

Version: 2 Date: 04 Apr 2019

Author’s response to reviews:

Dear editor(s) of BMC Medicine,

We sincerely thank you for giving us the opportunity to submit a revised version of our original manuscript entitled "Vedolizumab Trough Level Monitoring In Inflammatory Bowel Disease: A State Of The Art Overview" by Pouillon et al. (BMED-D-19-00029R1), for consideration in BMC Medicine.

We have done the necessary changes as requested. Included you can find a point-by-point response to the remarks.

Sincerely,

Lieven Pouillon

Point-by-Point response:

Comment 1: Please format the abstract according to our minireview guidelines (should be structured with a background, main body of the abstract and short conclusion).

Answer 1: We have adapted the abstract accordingly.

Comment 2: Please remove 'Anno 2019' from the title

Answer 2: We have removed this from the title.
Comment 3: Would you be able to please provide a higher quality figure? The current version is not easily readable

Answer 3: We have uploaded a higher quality figure. If necessary, I can also send you the figure in PowerPoint, which might allow you to further process it as required.